Author's response to reviews

Title: Transcription profile of non-immortalized breast cancer cell lines.

Authors:

Mariana Fernandez-Cobo (mariana.fernandez@mssm.edu)
James F Holland (james.holland@mssm.edu)
Beatriz GT Pogo (beatriz.pogo@mssm.edu)

Version: 7 Date: 29 March 2006

Author's response to reviews:

The changes were made according to the instructions below.

Abbreviations- Could you please place this section before the Competing interests section

Tables - Please choose one of the following options (1) refer to the 'up regulated' and 'down regulated' sections of the tables as individual tables (in consecutive numbering fashion i.e. up regulated = table 3; down regulated = table 4) or (2) include the tables split up in this manner as additional files or (3) include the headings within the table and structure the two separate sections in one complete table. Unfortunately the present layout cannot be processed by our computing systems. Please also remember to change the references to the tables in the manuscript text.